Author’s response to reviews

Title: Caretakers' Understanding of Malaria, Use of Insecticide Treated Net and Care Seeking-Behavior for Febrile Illness of their Children in Ethiopia

Authors:
Zewdie Birhanu (zbkoricha@yahoo.com)
Yemane Ye-ebiyo Yihdego (yemaneyy@gamil.com)
Delenasaw Yewhalaw (delenasawye@yahoo.com)

Version: 3 Date: 10 Sep 2017

Author’s response to reviews:

1. Please make sure the abstract follows the format outlined in our submission guidelines.-https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-articleCurrently, the "Methods" section is missing.Response: Thank you for the comment. The abstract is updated as per the guideline. “Methods” is added (page 2, line 17). Please make sure the submission data is also updated with any abstract changes.Response: all submission data were updated 2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.Response: The manuscript is uploaded as a single, final and clean version without any track changes or highlighted colors or comments. Figures (clean versions) were uploaded as separate files.